Author’s response to reviews

Title: Using information and communication technologies to involve patients and the public in health education in rural and remote areas: a scoping review

Authors:

Maman Joyce Dogba (joyce-maman.dogba@fmed.ulaval.ca)
Anara Dossa (richi.dossa.1@ulaval.ca)
Erik Breton Breton (Erik.Breton@fmed.ulaval.ca)
Ruth Gandonou-Migan (ruthtisam@yahoo.fr)

Version: 1 Date: 20 Nov 2017

Author’s response to reviews:

Dear Editor,

Please see the responses to your comments below from you/your team. We have additionally uploaded a cover letter to resubmit for consideration for publication in your journal. Please see the specific revisions made as follows:

Editor's comments:

1. Please include an email address for each author on the title page. - All emails of authors added to manuscript

2. Please amend/add to the abstract in your manuscript to include the following subheadings - Background: the context and purpose of the study

Methods: how the study was performed and statistical tests used

Results: the main findings

Conclusions: brief summary and potential implications

- Have updated subheadings in abstract to reflect above
3. Please include a statement in your Declarations to confirm that your study did not require ethics approval or participant consent. An example of a statement is “Ethics approval and participant consent was not necessary as this study involved the use of a previously-published de-identified database according to [details of national legislation]. - Updated declarations and in particular, added to the Ethics Approval section as requested

4. Please remove supplementary figure legend(s) so that they are not included in the main manuscript PDF. - Anything seeming to be a legend has been removed from the figured